Author's response to reviews

Title: "Parry Romberg syndrome with a wide range of ocular manifestations and some new findings (hypoplasia of the ciliary body): a case report."

Authors:

Antonio Maria Fea (antoniomfea@gmail.com)
Vittoria Aragno (vittoria.aragno@hotmail.it)
Cristina Briamonte (cristina.briamonte@gmail.com)
Mauro Franzone (maurofranzone@virgilio.it)
Davide Putignano (edivado461@gmail.com)
Federico Grignolo (federico.grignolo@unito.it)

Version: 2
Date: 18 April 2015

Author's response to reviews:

Dear Editor,
in this new revised submission we have:
- added the title page in the manuscript
- revised the abstract and manuscript sections
- added the acknowledgments
- removed the patient name from the figure 1
- added the checklist
Thank you